April 18, 2000

Math 103:99 The last homework assignment

Write a paper between one and two pages long about the Data Encryption Standard (DES) and its policy implications. The paper is due Wednesday, April 26. You may use

http://math.rutgers.edu/ greenfie/currentcourses/math103spring/des.html

(or go to the course web page and follow The DES assignment link). Topics are suggested there. Please cite any references. Be sure to explain any technical terms you use carefully with standard English words and sentences.

April 18, 2000

Math 103:99 The last homework assignment

Write a paper between one and two pages long about the Data Encryption Standard (DES) and its policy implications. The paper is due Wednesday, April 26. You may use

http://math.rutgers.edu/ greenfie/currentcourses/math103spring/des.html

(or go to the course web page and follow The DES assignment link). Topics are suggested there. Please cite any references. Be sure to explain any technical terms you use carefully with standard English words and sentences.